

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dovern, Jonas; Müller, Lena Sophia; Wohlrabe, Klaus

## Working Paper Local Information and Firm Expectations about Aggregates

CESifo Working Paper, No. 9826

**Provided in Cooperation with:** Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

*Suggested Citation:* Dovern, Jonas; Müller, Lena Sophia; Wohlrabe, Klaus (2022) : Local Information and Firm Expectations about Aggregates, CESifo Working Paper, No. 9826, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/263756

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# Local Information and Firm Expectations about Aggregates

Jonas Dovern, Lena Sophia Müller, Klaus Wohlrabe



### Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email office@cesifo.de Editor: Clemens Fuest https://www.cesifo.org/en/wp An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com

- from the RePEc website: <u>www.RePEc.org</u>
- from the CESifo website: <u>https://www.cesifo.org/en/wp</u>

# Local Information and Firm Expectations about Aggregates

## Abstract

Using new survey data on quantitative growth expectations of firms in Germany, we show that firms resort to local information when forming expectations about aggregate growth. Firms extrapolate from the economic situation in their county, industry growth and their individual business situation. The effect is particularly strong for small firms and explains part of the high expectation dispersion across firms. Furthermore, we show that growth expectations are correlated with employment and investment decisions of firms, highlighting that differences in expectations do indeed seem to lead to differences in actual firm decisions. Our results confirm predictions of theoretical models with rational inattention.

JEL-Codes: D840, E200, E320.

Keywords: GDP expectations, expectation heterogeneity, disagreement, rational inattention, ifo business tendency survey.

Jonas Dovern\* Friedrich-Alexander University Erlangen-Nürnberg, Lange Gasse 20 Germany – 90403 Nuremberg jonas.dovern@fau.de

Lena Sophia Müller Friedrich-Alexander University Erlangen-Nürnberg, Lange Gasse 20 Germany – 90403 Nuremberg lena.sophia.mueller@fau.de

Klaus Wohlrabe ifo Institute – Leibniz Institute for Economic Research at the University of Munich / Germany wohlrabe@ifo.de

\*corresponding author

July 1, 2022

We would like to thank the team of ifo's Economics and Business Data Center for excellent support and provision of the data. We also thank Tobias Lohse and Sebastian Link for sharing their Stata code. We are grateful to Mirko Wiederholt for the very valuable comments on an earlier draft. Jonas Dovern acknowledges financial support by Deutsche Forschungsgemeinschaft through SPP 1859 (project number DO 2296/1-1). This working paper is a completely revised version of CESifo Working Paper No. 8179 "How Do Firms Form Expectations of Aggregate Growth? New Evidence from a Large-Scale Business Survey".

#### 1 Introduction

Expectations play a central role in current macroeconomic models and it is widely believed that they are a key driver of aggregate fluctuations (Beaudry and Portier, 2007; Angeletos and La'O, 2013). Yet, it remains controversial how macroeconomic expectations are formed and how macroeconomists should model this process. In particular, evidence on how firms form macroeconomic expectations remains very scarce. As firms are of central importance for price setting as well as labor and investment demand, a better understanding of what determines their expectations is of crucial importance for macroeconomic research and policy. Recently, Andrade et al. (2022) provide empirical evidence that—in line with the "island" model proposed by Lucas (1972)—industry conditions affect firms' views of macroeconomic conditions. In this paper, we provide more evidence that local business conditions do indeed play a role when firms form expectations about aggregate economic variables in a way that is consistent with the predictions of a model with rational inattention as in Mackowiak and Wiederholt (2009).

In particular, we show that firms' expectations of the growth rate of the (real) gross domestic product (GDP), henceforth "growth expectations", are driven by economic conditions in the firms' headquarters county and in their industry as well as by the idiosyncratic business situation of firms. This means that the local signals that firms use to form their macroeconomic expectations are not limited to industry-specific information. On balance and consistent with models of rational inattention, these effects are larger for small firms that presumably do not have the capacity to allocate attention to aggregate conditions and thus away from idiosyncratic conditions (Maćkowiak and Wiederholt, 2009). We also show that those small firms—in contrast to large firms—do not act upon their growth expectations when it comes to employment and investment decisions, again supporting the idea of firms being inattentive, yet rational.

We obtain this evidence by augmenting the ifo Business Tendency Survey (BTS), a large-scale, high-level, and fairly representative business survey in Germany,<sup>1</sup> with a ques-

<sup>&</sup>lt;sup>1</sup>Sauer and Wohlrabe (2019) document that the BTS is usually answered by senior managers. As a consequence of the long-standing nature of the survey, young firms and start-ups are underrepresented.

tion that elicits growth expectations of firms in several survey waves during the period from 2018 to 2020. We can link these expectations about aggregate growth to i) other firm-level information from the survey (such as business expectations, business decisions, and firm size), ii) local unemployment (which we use as a proxy for regional economic conditions), and iii) industry-specific information. This allows us to estimate the nexus between local information and macroeconomic expectations using panel regressions that feature a wide range of fixed effects to control for unobserved confounders. To mitigate potential endogeneity problems, we use carefully selected subsamples of firms whose local variables are unlikely to be driven by their growth expectations (e.g., because their business expectations were noncyclical in the past). The large (roughly 4, 800 firms) and fairly representative sample of firms in Germany makes it ideal for studying heterogeneity in expectation formation across different types of firms.

We study growth expectations—thereby complementing other studies that focus on inflation expectations of firms—because aggregate demand is a key variable for fluctuations in corporate investment demand (Bachmann and Zorn, 2020). Since GDP is the broadest measure of aggregate demand available, firms' expectations of GDP growth are an important driver of their business decisions (Tanaka et al., 2020).<sup>2</sup>

We first document that the heterogeneity of growth expectations across firms is large, indeed more similar to that observed for private households than for professional macroeconomic forecasters. This is in line with evidence in Coibion et al. (2018); Tanaka et al. (2020) and Candia et al. (2021, 2022). In all economic sectors that we investigate (manufacturing, trade, and services), the dispersion of growth expectations is higher for small firms than for large firms. Heterogeneity is also higher for firms whose business does not depend on the general business cycle in Germany. These findings support theories of rational inattention that imply that firms pay less attention to economic information if the fixed cost of doing so are large relative to the size of the firm and/or the benefits of closely monitoring the business cycle are low.

 $<sup>^{2}</sup>$ Even if sector-specific demand is eventually what firms are interested in, the demand for specific products is likely to depend strongly on aggregate demand in most industries.

We then turn to the question of whether information that firms gather in their local environment ("island") shape their expectations for aggregate growth and, thus, partially explain the heterogeneity of macroeconomic expectations. We find that firms which are located in a county that is not doing well economically (as indicated by high local unemployment) tend to have more pessimistic growth expectations. A 1 percentage point increase of the unemployment rate in the home county of a firm is associated, on average, with roughly 31 basis points lower growth expectations of this firm. We observe a similar—albeit very small—effect from industry-specific economic conditions on firms' aggregate growth expectations. Finally, we find that firms which report a positive current business situation, positive expectations for their own business or expectations of rising product prices report more optimistic growth expectations. Overall, these effects are driven by small firms in our sample while growth expectations of larger firms are more independent of local information.

The evidence that firms extrapolate from their own economic experiences to the broader national business cycle situation is in line with "island models" (Lucas, 1972) that have recently been picked up by the modern macroeconomic literature on models with rational expectations and information frictions (e.g. Lorenzoni, 2009; Angeletos and La'O, 2013; Nimark, 2014). The fact that small firms pay more attention to local information when forming expectations of aggregate variables is consistent with models of rational inattention which postulate that firms need to optimally allocate a limited amount of attention. It is plausible that small firms have less attention at disposal overall (Maćkowiak and Wiederholt, 2009). Our evidence complements several studies that show in other contexts how "local" information significantly affects macroeconomic expectations and is an important driver of expectation heterogeneity (Berger et al., 2009, 2011; Kuchler and Zafar, 2019; Andrade et al., 2022).

Finally, we document that growth expectations are positively related to firms' investment and labor demand. Even after controlling for reported business situation and expectations (and a number of fixed effects) higher growth expectations are associated with a higher propensity of firms to increase investment spending and the number of employees. This complements evidence in Coibion et al. (2020) who show that inflation expectations of firms affect business decisions. Looking at heterogeneity across the size distribution of firms, we find that the effect is driven by large firms while small firms do not resort to their growth expectations when making employment and investment decisions. This suggest that small firms are aware that their growth expectations are extrapolated from local information and do not act upon them. Or in other words: small firms are more likely (relative to large firms) to rationally choose not to invest into acquiring information beyond what they notice "on their islands" when forming expectations of aggregates and, at the same time, are wise enough to not rely too much on those macroeconomic expectations when making business decisions.

From a broader perspective, our paper contributes to a booming literature that studies how market participants—most importantly private households and firms—form macroeconomic expectations (Manski, 2018). Contributions by, inter alia, Manski (2004), Mankiw and Reis (2002), Sims (2003), Woodford (2003), and Coibion and Gorodnichenko (2015) have spurred a rapidly growing theoretical and empirical literature that aims to measure and model macroeconomic expectations more realistically and more coherently with properties of observed macroeconomic expectations than the FIRE model.

The empirical evidence on how firms form expectations yet remains scarce as many surveys collect only qualitative data or are limited to specific subsets of firms based on a firm's sector or size (Andrade et al., 2022; Tanaka et al., 2020).<sup>3</sup> Kumar et al. (2015) and Coibion et al. (2018) analyze a more representative sample of firms in New Zealand. Candia et al. (2021) use a monthly panel of survey-based inflation expectations of US firms to show that the size and in particular the industry of a firm affects expectations. Link et al. (2021) combine data from the BTS with a survey among private households to compare firm expectations with those of households. With the exception of Tanaka et al. (2020), these papers primarily focus inflation expectations. Candia et al. (2022) provide an overview of the literature on inflation expectations of firms.

<sup>&</sup>lt;sup>3</sup>The literature on household expectations is more comprehensive. Still, only few data sets contain information on quantitative expectations of private households. Exceptions include Bruine De Bruin et al. (2011), Das et al. (2020), Malmendier and Nagel (2011), and Malmendier and Nagel (2016).

The remainder of this paper is structured as follows. Section 2 describes the data that we use, in particular the novel data on quantitative corporate growth expectations. Section 3 presents evidence on the expectation formation process of firms, which includes details on the dispersion of expectations (Section 3.1), results on the impact of local conditions on expectations (Section 3.2), and a review of the consistency of our findings with theoretical macroeconomic models (Section 3.3). Section 4 analyzes the effect of growth expectations on firms' employment and investment decisions. Section 5 concludes.

#### 2 Data and Descriptive Statistics

Most of our data come from the ifo Business Tendency Survey (BTS) conducted monthly by the ifo Institute. The survey covers various business aspects for a panel of firms.<sup>4</sup> We use data on firms from the manufacturing sector, the trade sector, and the service sector. For about 98 % of firms, one particular person is responsible for regularly answering the questionnaire (Sauer and Wohlrabe, 2019). More than 80 % of those persons are in an upper management position such as owner, CEO, or department head. This suggests that respondents have a good overview of their company and provide answers that accurately reflect sentiments representative of the firm.

We elicit quantitative growth expectations in the survey waves conducted in August 2018, March and August 2019 and August 2020. We asked firms to report their expected annual growth rate of real GDP for the current and next year with one decimal.<sup>5</sup> This results in expectations data for four target years collected at four time points. On average, about 80 % of firms that returned a questionnaire answered our question, leaving us with about 4,500 to 5,000 observations per survey wave.<sup>6</sup>

<sup>&</sup>lt;sup>4</sup>The BTS is technically conducted at the product level, i. e., some larger firms answer more than one questionnaire. We aggregate the answers to the firm level by using the mean over all questionnaires returned by one firm as proposed by Link (2020). For qualitative questions we transform the mean of all questionnaires returned by one firm back into discrete answer categories.

<sup>&</sup>lt;sup>5</sup>See Appendix B for the exact wording of the special questions. In the wave of August 2018, we asked about growth expectations for the current calender year only.

 $<sup>^{6}</sup>$ We drop all observations that lie outside a three-standard-deviations interval around the median growth expectation for each combination of survey wave and target year to exclude unreasonably high or low expectations. Overall, we lose about 1 % of the observations.

| Survey      | Aug '18 | Mar '19 |       | Aug '19 |       | Aug '20 |       |
|-------------|---------|---------|-------|---------|-------|---------|-------|
| Expectation | 2018    | 2019    | 2020  | 2019    | 2020  | 2020    | 2021  |
| Ν           | 4,641   | 4,831   | 4,774 | 4,856   | 4,831 | 5,010   | 4,961 |
| Mean        | 1.8     | 1.2     | 1.1   | 0.9     | 0.8   | -6.2    | 1.5   |
| Median      | 1.8     | 1.2     | 1.0   | 1.0     | 1.0   | -7.5    | 2.0   |
| Std. Dev    | 1.3     | 1.4     | 1.5   | 1.6     | 1.7   | 7.9     | 6.2   |

Table 1: Descriptive Statistics for Quantitative Growth Expectations

**Notes**: All numbers refer to the entire sample after dropping any observations that exceed the median by three standard deviations in each wave. In total, we drop about 1 % of all answers.

In the first three survey waves, average growth expectations range from 0.8 % to 1.8 % which is slightly higher than the actual GDP growth rates in those years (Table 1). In the survey from August 2020 average expectations for the same year drop sharply in response to the COVID-19 pandemic. A similar disruption is visible when considering the dispersion of expectations. Standard deviations increase slightly from 1.3 % to 1.7 % between the surveys in 2018 and 2019, followed by a stark rise beyond 6 % in August 2020. We do not observe strong differences in average growth expectations across sectors (Table A.1 in Appendix A).

We link three different types of local information to growth expectations of firms and further use these information to compute historical correlations of local economic conditions with the German business cycle to construct subsamples of "acyclical" firms.

First, we use local unemployment rates to approximate local business cycle conditions. These data are provided by the German Federal Employment Agency (Bundesagentur für Arbeit, BA). The monthly observations for 401 counties cover the period from 1991 to August 2020. To eliminate the effect of any seasonal movements in unemployment which tend to differ substantially across counties, we consider twelve-months moving averages of unemployment rates. Second, we measure industry conditions by the (quarterly) growth rates of sales or production in 379 industries. The data are from the German Federal Statistical Office (Statistisches Bundesamt). Finally, we use three measures from the regular BTS to measure firm-specific idiosyncratic business assessment, firms' business expectations, and firms' expectations of their own product prices (all reported on a trichotomous scale from -1 (bad/decreasing) to 1 (good/increasing)).

To analyze heterogeneity across the size distribution of firms, we construct a measure of firm size based on information from the BTS. Unfortunately, the survey does not provide a harmonized variable for firm size because the questionnaires differ across sectors. Therefore, we create a categorical measure of firm size with values ranging from 1 (very small) to 5 (very large). In the manufacturing and trade sector the measure is based on the number of employees, in the service sector it is based on turnover.

Finally, we use self-reported and qualitative assessments of planned and realized changes in employment and investment from the BTS to identify potential effects of growth expectations on business decisions. Again, these variables are reported on a trichotomous scale from -1 (decrease) to 1 (increase). The forward-looking measures refer to planned changes in the number of employees over the next two to three months and planned changes in investment expenditure in the current calendar year (relative to the previous year), respectively. The backward looking measures refer to employment changes over the past two or three months (depending on the sector) and the change in investment volume in the last calender year, respectively. In terms of timing, we match each expectation wave with the earliest available observation of the forward-looking measures (employment plans are covered by the survey every month, investment plans only in May and November). In the case of backward looking measures, we match each expectation wave with the assessment of employment from two or three months later (depending on the sector) and with the assessment of investment from the next calendar year, respectively.

In addition, we control for the historical volatility of business expectations, a firm's success in predicting their own business situation in the past and whether a firm answered the questionnaire online or on paper.<sup>7</sup>

To construct subsamples of firms that are relatively independent of the German business cycle we use five different approaches. First, we select firms based on the historical correlation of their business assessment with GDP growth over the period from 1990 to 2020. Second, we select those firms that make a large share of their turnover on export

<sup>&</sup>lt;sup>7</sup>We measure historical volatility by the standard deviation of the monthly responses between 1991 and 2018. We compute a firm's success in predicting its own business situation by the firm-specific expectation error as in Bachmann and Elstner (2015).



Figure 1: Dispersion of Growth Expectations for Different Waves and Target Years.

**Notes:** This figure shows histograms of firm's GDP expectations for the years 2018 to 2021 elicited in four different survey waves. The red lines indicate the mean expectations.

markets. We do this based on self-reported export shares elicited using a special survey question in September 2018. Third, we use answers to a special question in August 2018, which asked firms to indicate on a five-point scale how dependent their business model is on the German business cycle ("GDP importance"). The fourth sub-sample includes firms in counties where the correlation between local unemployment and German GDP growth historically has been low. Lastly, we select companies in industries with low correlation between industry growth and general economic growth. Table A.2 in Appendix A contains a set of descriptive statistics for all variables.

#### 3 Heterogeneity of Corporate Growth Expectations

#### 3.1 Expectation Dispersion and Firm Size

It is a well known fact that macroeconomic expectations are dispersed if compared across individuals (e.g., Mankiw et al., 2003; Dovern et al., 2012). Similar to previous findings by, inter alia, Coibion et al. (2018) and Tanaka et al. (2020) we find that the dispersion of firms' growth expectations is large.

The histograms in Figure 1 show a very broad distribution for each survey wave and target year, especially during the COVID-19 pandemic in 2020. The cross-sectional

| Panel A: if  | o BTS and     | Professi     | onal Forec   | asters           |            |              |              |              |
|--------------|---------------|--------------|--------------|------------------|------------|--------------|--------------|--------------|
|              | ifo E         | BTS          | Consens      | Consensus Econ.  |            | SPF (ECB)    |              | (Fed)        |
| Target       | Survey        | Std.<br>Dev. | Survey       | Std.<br>Dev.     | Survey     | Std.<br>Dev. | Survey       | Std.<br>Dev. |
| Same Year    | Aug. 18       | 1.3          | Aug. 18      | 0.1              | Oct. 18    | 0.1          | Aug. 18      | 0.1          |
| Same Year    | Mar. 19       | 1.4          | Mar. 19      | 0.3              | Apr. 19    | 0.2          | Feb. 19      | 0.2          |
| Next Year    | Mar. 19       | 1.5          | Mar. 19      | 0.3              | Apr. 19    | 0.3          | Feb. 19      | 0.5          |
| Same Year    | Aug. 19       | 1.6          | Aug. 19      | 0.2              | Oct. 19    | 0.1          | Aug. 19      | 0.1          |
| Next Year    | Aug. 19       | 1.7          | Aug. 19      | 0.4              | Oct. 19    | 0.2          | Aug. 19      | 0.5          |
| Same Year    | Aug. 20       | 7.9          | Aug. 20      | 0.7              | Oct. 20    | 0.6          | Aug. 20      | 0.8          |
| Next Year    | Aug. 20       | 6.2          | Aug. 20      | 1.2              | Oct. 20    | 0.8          | Aug. 20      | 2.5          |
| Panel B: if  | o BTS and     | Other fi     | rm surveys   | 5                |            |              |              |              |
|              | ASBC          | Japan        | Firm Su      | urvey NZ         |            |              |              |              |
| Target       | Survey        | SD           | Survey       | SD               |            |              |              |              |
| 4q ahead     | Various       | 1.3*         | Various      | $0.51.0^\dagger$ |            |              |              |              |
| Notes: Measu | ires from the | ifo BTS      | refer to the | trimmed se       | ample * Ta | naka et a    | 1 (2020) rep | ort the av   |

Table 2: Dispersion of Growth Expectations in Different Surveys

**Notes**: Measures from the ifo BTS refer to the trimmed sample. \* Tanaka et al. (2020) report the average standard deviation across all survey waves from 1989 to 2015. <sup>†</sup> The survey was conducted four times and the standard deviations of growth expectations where 0.7, 1.0, 0.5, and 0.6 percentage points, respectively.

standard deviation of growth expectations of German firms is much higher than those of professional forecasters (Panel A in Table 2). This is in line with findings for inflation expectations of firms (Andrade et al., 2022; Candia et al., 2022). The cross-sectional standard deviation in our sample (before the pandemic) is very similar to that observed for growth expectations of firms in Japan and New Zealand (Tanaka et al., 2020; Coibion and Gorodnichenko, 2015) (Panel B of Table 2).<sup>8</sup> Overall, the large disagreement indicates that growth expectations are formed based on different and potentially noisy signals about the state of the economy that firms are exposed to. We analyze the role of local information in that context in Section 3.2.

Figure 2 shows, for each sector, how the cross-sectional standard deviation of growth expectations varies with firm size. There is a clear downward trend in dispersion with increasing firm size. We observe the same trend when using the categorical measure that we use to measure firm size homogeneously across the three sectors (Appendix A,

<sup>&</sup>lt;sup>8</sup>Due to the lack of surveys that contain information about quantitative growth expectations of households a comparison to corresponding results for private households is not possible. Link et al. (2021) document lower levels of dispersion among firms than among households for expectations of inflation, unemployment, and the policy rate.





**Notes:** The graph shows the standard deviation of firms' GDP growth expectations according to their size from small to large. For the manufacturing sector and the trade sector, we measure firm size by the number of employees and calculate the standard deviation over rolling windows of 100 firms after ordering the firms by firm size. In the service sector we calculate standard deviations for each turnover category available in the BTS.

Figure A.1). The differences in standard deviations across the five size categories are significantly different from zero (Appendix A, Table A.3).<sup>9</sup>

Overall, we find that growth expectations of firms are much more dispersed than those of professional forecasters and that the level of dispersion varies across the size distribution of firms. We now turn to the question of whether differences in local information partly explain the heterogeneity of expectations.

#### 3.2 Local Conditions and Growth Expectations

To empirically analyze the link between local information and expectations of aggregate growth, we resort to local unemployment (as a proxy for the local business situation), industry growth (as a proxy for the economic situation of individual industries), business assessment and expectations of future product prices (as proxies for idiosyncratic information and shocks they face). Our focus is the analysis of the cross-sectional variation. We remove common time effects on expectations by adding fixed effects for each survey wave and target year in all specifications ("Wave FE"). Since quantitative aggregate growth

<sup>&</sup>lt;sup>9</sup>The pattern is less pronounced when we split the sample according to the self-reported measure for the importance of the business cycle (Panel B of Figure A.1 and Table A.4 in Appendix A).

expectations were collected in only four waves of the BTS, we cannot analyze how growth expectations adjust dynamically to local shocks like Andrade et al. (2022) do.

For all analyses in this section, the dependent variable is growth expectations of firms. We add regional and industry fixed effects to capture variation in growth expectations driven by unobserved local or industry conditions.<sup>10</sup> In additional regressions, we add control variables measuring firm-specific characteristics (firm size, historical volatility of business expectations, success in predicting the own business situation, answering online or on paper). Finally, we estimate models with firm fixed effects to capture any firmspecific unobservables. In these regressions, the link between local information and growth expectations is identified based on within-firm variation.

To address potential heterogeneity in the strength of extrapolation from local conditions across the firm-size distribution, we also consider specifications with interaction terms. The interaction terms refer to the categorical measure of firm size, which is consistent across sectors.

To minimize the risk to find correlations between growth expectations and local conditions only because the latter might be driven by the former (a mechanism that would be diametrically opposed to the one suggested by "island" models), we look at subsamples of firms for which this is unlikely. We follow Andrade et al. (2022) and select firms from regions or industries that historically have been very acyclic. In the case of the idiosyncratic business situation, we base the subsamples on various measures of a firm's dependence on the German business cycle.

The descriptive results above suggest that the COVID-19 pandemic had strong effects on expectation disagreement among firms. It is very likely that the expectation formation process was heavily influenced by the pandemic in 2020 and quite different from normal times. For this reason, we estimate some specifications using samples that exclude data collected in 2020.

 $<sup>^{10}</sup>$ The firms in our sample cover all of the 401 administrative districts, called "Kreise und kreisfreie Städte". The 379 industries are defined by the German standard classification system of 2008 (*WZ 08*).

#### 3.2.1 Local Economic Environment

The first local measure is the economic situation firms experience in the area where they do business. The idea is that firms generalize from the conditions in their county to the business cycle situation of the entire country since they confound local and aggregate shocks. Given that each individual county is too small to have a substantial impact on the overall growth rate in Germany, such behavior would constitute a deviation from optimal forecasting (under full information) and could explain some of the variation of expectations across firms.

We use the local unemployment rate to measure the strength of the local economy because it is one of the few economic indicators that is available without major publication lag—and, hence, observable for the firms in near real time.<sup>11</sup> To measure the relative economic strength of a county, we simply take the average of the unemployment rate during the 12 months ending with the respective survey wave.

Table 3 displays the results. Columns 1-4 show the results when gradually including more fixed effects and controls. Beginning with column 2 where we include control variables, region and industry fixed effects in addition to wave fixed effects, the coefficient has the expected sign and is statistically significant at the 5 % level. A one percentage point increase in the unemployment rate (in a county) is associated with growth expectations (for Germany) of local firms that are 0.3 percentage points lower. This effect is conditional on the current economic situation in Germany for which we control with wave-target year fixed effects. We find equivalent results when using firm fixed effects (column 3). Firms that experience an increase in local unemployment adjust expectations about aggregate growth downward.

Column 4 shows the estimates for different firm sizes. The baseline corresponds to the smallest firm size category; the interaction terms quantify the additional effects for firms in the other four size categories in ascending order. Summing up the baseline and the respective interaction coefficient yields the total effects for these larger companies.

 $<sup>^{11}\</sup>mathrm{We}$  use information about the firms' ZIP codes and the municipality names in their address to assign them to counties. We loose approximately 11 % of answers because either no information about the location is provided or because – due to reporting errors – the reported combination of ZIP code and municipality name does not allow identifying an unambiguous county.

The baseline coefficient is highly significant and suggests a stronger effect for small firms compared to the overall effect in the previous column. The interaction coefficients increase with firm size, suggesting a fading impact of unemployment on expectations for larger firms. In fact, large firms (size category four and five) do not extrapolate from local conditions. The total effect for those firms is not statistically different from zero. In the pre-pandemic sample, only the baseline coefficient stays significant (column 5).

Focusing on firms in acyclical counties in column 6, the baseline coefficient is significant and higher compared to the full sample (column 4). This implies that local conditions inform growth expectations of small firms more in acyclical counties. Again, the total effects for larger firms are not significantly different from zero. We do not find significant results when excluding the survey wave from 2020 in the last column.

To quantify the amount of dispersion in expectations attributed to the local economic environment, we compare the explained share of variation  $(R^2)$  of a model with merely wave-target year fixed effects to that of a model with controls for the local economic environment. The comparison is based on data collected before 2020 to exclude effects of the greatly increased variance due to the COVID-19 pandemic. Local unemployment raises the explained share of variation in expectations by 0.5 %. When including county fixed effects instead of local unemployment, we observe an increase of 72.3 %.

#### 3.2.2 Industry Environment

Firms might also confuse industry conditions with the state of the aggregate economy. Andrade et al. (2022) show empirically that expectations of French manufacturing firms about prices and production react fast to industry-specific shocks even if these shocks do not affect aggregate conditions.

We match annualized quarterly log growth rates in 379 industries to the survey data and perform regressions similar to the regional case with industry growth as the main explanatory variable.<sup>12</sup>

<sup>&</sup>lt;sup>12</sup>We retrieve quarterly industry-specific data from the German Federal Statistical Office. Growth rates for industries in the trade sector are based on sales, growth rates for industries in the manufacturing and service sector are based on production.

|                           |         |          | All counties | 3             |                | Acyclica     | l counties     |
|---------------------------|---------|----------|--------------|---------------|----------------|--------------|----------------|
|                           |         |          |              |               | $\rm w/o~2020$ |              | $\rm w/o~2020$ |
|                           | (1)     | (2)      | (3)          | (4)           | (5)            | (6)          | (7)            |
| U                         | 0.025** | -0.304** | -0.310**     | -0.421***     | -0.228**       | -0.717**     | -0.138         |
|                           | (0.011) | (0.134)  | (0.133)      | (0.157)       | (0.102)        | (0.365)      | (0.205)        |
| U $\times$ Firm size 2    |         |          |              | 0.025         | -0.010         | 0.352        | -0.035         |
|                           |         |          |              | (0.108)       | (0.050)        | (0.319)      | (0.139)        |
| U $\times$ Firm size 3    |         |          |              | 0.122         | -0.099         | 1.139***     | -0.052         |
|                           |         |          |              | (0.148)       | (0.070)        | (0.412)      | (0.205)        |
| $\rm U\times Firm$ size 4 |         |          |              | $0.274^{*}$   | -0.107         | $0.956^{**}$ | -0.242         |
|                           |         |          |              | (0.162)       | (0.078)        | (0.431)      | (0.224)        |
| U $\times$ Firm size 5    |         |          |              | $0.594^{***}$ | -0.078         | $1.167^{**}$ | -0.287         |
|                           |         |          |              | (0.180)       | (0.086)        | (0.499)      | (0.240)        |
| Wave FE                   | Yes     | Yes      | Yes          | Yes           | Yes            | Yes          | Yes            |
| Region FE                 | No      | Yes      | No           | No            | No             | No           | No             |
| Industry FE               | No      | Yes      | No           | No            | No             | No           | No             |
| Firm FE                   | No      | No       | Yes          | Yes           | Yes            | Yes          | Yes            |
| Controls                  | No      | Yes      | No           | No            | No             | No           | No             |
| N                         | 29,366  | 28,339   | 29,366       | 29,366        | 21,231         | 5,874        | 4,255          |
| $R^2$                     | 0.30    | 0.34     | 0.37         | 0.37          | 0.09           | 0.33         | 0.09           |

Table 3: Impact of Local Unemployment on Growth Expectations

*Notes:* Columns 6-7 display estimates for subsamples of firms in counties with low historic correlation between local unemployment and German GDP growth. \*p<0.1, \*\*p<0.05, \*\*\*p<0.01. Standard errors clustered at county level in parentheses.

The coefficient on industry growth in columns 1-3 of Table 4 is highly significant and stable across all three specifications. Economically however, the effect is small: a 1 % increase in industry growth is related to an upward adjustment of expectations by 0.004 percentage points.

Mainly small firms are behind the effect as the baseline coefficient in column 4 shows. The interaction terms do not indicate a clear pattern and are mostly insignificant. Only firms in size category 4 do not seem to extrapolate from industry conditions. For those firms, the total effect is not statistically different from zero. We do observe a more distinct difference between small and larger firms in the sample of firms in acyclical industries (column 6). While small firms extrapolate from industry conditions (the baseline coefficient is now 0.02), larger firms in size category 2, 4 and 5 do not. Firms in size category 3 seem to step out of line with a positive (though insignificant) interaction coefficient.

In sum, also industry conditions explain some of the variation in growth expectations. In the pre-COVID-19 sample, industry growth increases the  $R^2$  by about 0.3 % compared to a model with only wave-target year fixed effects. Including industry fixed effects leads to an increase by 69.4 %.

|                         |           |           | All industries |           |                | Acyclical | industries |
|-------------------------|-----------|-----------|----------------|-----------|----------------|-----------|------------|
|                         |           |           |                |           | $\rm w/o~2020$ |           | w/o $2020$ |
|                         | (1)       | (2)       | (3)            | (4)       | (5)            | (6)       | (7)        |
| Industry growth         | 0.0039*** | 0.0037*** | 0.0037***      | 0.0036*** | 0.0056**       | 0.0192**  | 0.0016     |
|                         | (0.0004)  | (0.0005)  | (0.0005)       | (0.0012)  | (0.0025)       | (0.0080)  | (0.0039)   |
| IG $\times$ Firm size 2 |           |           |                | 0.0001    | -0.0011        | -0.0129   | 0.0046     |
|                         |           |           |                | (0.0013)  | (0.0031)       | (0.0091)  | (0.0045)   |
| IG $\times$ Firm size 3 |           |           |                | 0.0022    | 0.0042         | 0.0072    | 0.0029     |
|                         |           |           |                | (0.0016)  | (0.0043)       | (0.0134)  | (0.0073)   |
| IG $\times$ Firm size 4 |           |           |                | -0.0043** | -0.0016        | -0.0398** | 0.0011     |
|                         |           |           |                | (0.0019)  | (0.0055)       | (0.0180)  | (0.0094)   |
| IG $\times$ Firm size 5 |           |           |                | 0.0002    | -0.0015        | -0.0204   | 0.0022     |
|                         |           |           |                | (0.0018)  | (0.0044)       | (0.0134)  | (0.0070)   |
| Wave FE                 | Yes       | Yes       | Yes            | Yes       | Yes            | Yes       | Yes        |
| Region FE               | No        | Yes       | No             | No        | No             | No        | No         |
| Industry FE             | No        | Yes       | No             | No        | No             | No        | No         |
| Firm FE                 | No        | No        | Yes            | Yes       | Yes            | Yes       | Yes        |
| Controls                | No        | Yes       | No             | No        | No             | No        | No         |
| N                       | 32,067    | 27,243    | 32,067         | 32,067    | 22,650         | 7,146     | 5,014      |
| $R^2$                   | 0.30      | 0.34      | 0.39           | 0.39      | 0.09           | 0.37      | 0.08       |

Table 4: Impact of Industry Growth on Growth Expectations

*Notes:* Columns 6-7 display estimates for subsamples of firms in industries uncorrelated with German GDP growth. \*p<0.1, \*\*p<0.05, \*\*\*p<0.01. Standard errors clustered at county level in parentheses.

#### 3.2.3 Idiosyncratic Business Situation

Finally, firms might see "private information" about their own business situation as a signal about aggregate growth. They could (unconsciously) assume that a positive outlook for their own business translates to the aggregate economy. Similarly, a firm might interpret expectations about the future development of own sales prices not only as a signal about the demand for its own products but as informative for aggregated demand in the whole economy. To explore this link, we relate the quantitative growth expectations of firms to their reported business expectations and expectations about future product prices.

We face a problem of potential reverse causality at this point because a firm might well report pessimistic business expectations *because* it is pessimistic about future GDP growth. We address this concern in two ways while acknowledging that a proper identification of causal effects is hard given the data that we have at our disposal.

First, we argue that the issue is less problematic when looking at the relation between the assessment of the *current business situation* and growth expectations since the current situation of a firm depends on actual current factors rather than the expectation that a

|                                | Full s  | ample   | Importa      | nce $\geq 4$ | Foreign     | Sales $>75$   | GDP co      | orr. $(q_{0.1})$ | Industr.     | corr. $(q_{0.2})$ |
|--------------------------------|---------|---------|--------------|--------------|-------------|---------------|-------------|------------------|--------------|-------------------|
|                                | (1)     | (2)     | (3)          | (4)          | (5)         | (6)           | (7)         | (8)              | (9)          | (10)              |
| Bus. ass.                      | 0.29*** | 0.35*** | $0.51^{***}$ | $1.00^{**}$  | $0.50^{**}$ | 3.30***       | $0.52^{**}$ | $1.98^{***}$     | $0.54^{***}$ | $0.55^{**}$       |
|                                | (0.05)  | (0.11)  | (0.19)       | (0.39)       | (0.24)      | (1.02)        | (0.21)      | (0.50)           | (0.12)       | (0.25)            |
| Bus. ass. $\times$ Firm size 2 |         | -0.13   |              | $-1.14^{**}$ |             | $-3.64^{***}$ |             | $-1.59^{***}$    |              | -0.03             |
|                                |         | (0.12)  |              | (0.48)       |             | (1.08)        |             | (0.57)           |              | (0.30)            |
| Bus. ass. $\times$ Firm size 3 |         | -0.03   |              | 0.14         |             | $-2.37^{**}$  |             | -1.05            |              | 0.00              |
|                                |         | (0.15)  |              | (0.56)       |             | (1.09)        |             | (0.67)           |              | (0.41)            |
| Bus. ass. $\times$ Firm size 4 |         | 0.01    |              | -0.32        |             | -2.98**       |             | -2.08**          |              | 0.21              |
|                                |         | (0.18)  |              | (0.78)       |             | (1.18)        |             | (0.84)           |              | (0.47)            |
| Bus. exp. $\times$ Firm size 5 |         | -0.02   |              | -0.59        |             | -2.84**       |             | -2.81***         |              | -0.08             |
| -                              |         | (0.17)  |              | (0.82)       |             | (1.13)        |             | (0.73)           |              | (0.49)            |
| Wave FE                        | Yes     | Yes     | Yes          | Yes          | Yes         | Yes           | Yes         | Yes              | Yes          | Yes               |
| Region FE                      | No      | No      | No           | No           | No          | No            | No          | No               | No           | No                |
| Industry FE                    | No      | No      | No           | No           | No          | No            | No          | No               | No           | No                |
| Firm FE                        | Yes     | Yes     | Yes          | Yes          | Yes         | Yes           | Yes         | Yes              | Yes          | Yes               |
| Controls                       | No      | No      | No           | No           | No          | No            | No          | No               | No           | No                |
| N                              | 33,766  | 33,766  | 1,960        | 1,960        | 1,033       | 1,033         | 2,067       | 2,067            | 7,163        | 7,163             |
| $R^2$                          | 0.38    | 0.38    | 0.38         | 0.38         | 0.40        | 0.41          | 0.37        | 0.38             | 0.38         | 0.38              |

Table 5: Impact of Business Assessment on Growth Expectations

*Notes:* Columns 3-10 display estimates for subsamples consisting of firms whose business is independent of the German business cycle. We focus on firms that report that the German business cycle is unimportant for their business (columns 3-4), firms that make more than 75 % of their sales abroad (columns 5-6), firms with low historic correlation between their business assessment and German GDP growth in the past (columns 7-8), and firms in industries uncorrelated with German GDP growth (columns 9-10). \*p<0.1, \*p<0.05, \*\*\*p<0.01. Standard errors clustered at county level in parentheses.

firm has for the general economic outlook. Hence, we will regress growth expectations on the reported business assessment.

Second, we use subsamples of firms for which the German business cycle is not very important, i.e., firms that face demand that is not strongly correlated with the German business cycle. For those firms their expectation of domestic GDP growth should not affect their business outlook. A significant correlation would indicate that when forming growth expectations firms extrapolate from their business expectations. We select these subsamples in four ways. We rely on i) a self-reported measure of importance of the business cycle for the business of a firm, ii) information about export shares that firms reported in September 2018, iii) the historical correlation between German GDP growth and the business assessment of firms, and iv) firms in industries that are decoupled from aggregate growth as measured by the historical correlation between industry growth and aggregate growth. For each subsample, we estimate the overall effect as well as specifications that include interaction terms based on the firm size categories. All models include firm fixed effects.

Table 5 presents the results for our first strategy using business assessment as the main explanatory variable. The estimate in column 1 is highly significant and suggest

|                                | Full s                 | ample                  | Importa                | $nce \ge 4$            | Foreign        | Sales $>75$           | GDP c                | orr. $(q_{0.1})$        | Industr.       | corr. $(q_{0.2})$       |
|--------------------------------|------------------------|------------------------|------------------------|------------------------|----------------|-----------------------|----------------------|-------------------------|----------------|-------------------------|
|                                | (1)                    | (2)                    | (3)                    | (4)                    | (5)            | (6)                   | (7)                  | (8)                     | (9)            | (10)                    |
| Bus. exp.                      | $0.29^{***}$<br>(0.05) | $0.45^{***}$<br>(0.10) | $0.50^{***}$<br>(0.18) | $0.91^{**}$<br>(0.36)  | 0.13<br>(0.23) | 1.36<br>(0.85)        | $0.33^{*}$<br>(0.19) | $2.19^{***}$<br>(0.46)  | 0.13<br>(0.11) | $0.80^{***}$<br>(0.23)  |
| Bus. exp. $\times$ Firm size 2 | ( )                    | -0.14 (0.12)           | ( )                    | $-0.89^{**}$<br>(0.44) | ( )            | $-1.70^{*}$<br>(0.91) | ( )                  | $-2.27^{***}$<br>(0.54) | ( )            | $-0.87^{***}$<br>(0.28) |
| Bus. exp. $\times$ Firm size 3 |                        | $-0.38^{**}$<br>(0.16) |                        | -0.20<br>(0.55)        |                | -0.98<br>(1.01)       |                      | $-1.96^{***}$<br>(0.62) |                | $-1.27^{***}$<br>(0.38) |
| Bus. exp. $\times$ Firm size 4 |                        | 0.02<br>(0.18)         |                        | 0.27                   |                | -0.87<br>(1.12)       |                      | $-2.35^{***}$<br>(0.72) |                | -0.30<br>(0.42)         |
| Bus. exp. $\times$ Firm size 5 |                        | $-0.35^{*}$<br>(0.18)  |                        | -0.48<br>(0.83)        |                | -1.14 (1.01)          |                      | $-2.38^{***}$<br>(0.73) |                | $-0.89^{*}$<br>(0.48)   |
| Wave FE                        | Yes                    | Yes                    | Yes                    | Yes                    | Yes            | Yes                   | Yes                  | Yes                     | Yes            | Yes                     |
| Region FE                      | No                     | No                     | No                     | No                     | No             | No                    | No                   | No                      | No             | No                      |
| Industry FE                    | No                     | No                     | No                     | No                     | No             | No                    | No                   | No                      | No             | No                      |
| Controls                       | No                     | No                     | No                     | No                     | No             | No                    | No                   | No                      | No             | No                      |
| Firm FE                        | Yes                    | Yes                    | Yes                    | Yes                    | Yes            | Yes                   | Yes                  | Yes                     | Yes            | Yes                     |
| Ν                              | 33,725                 | 33,725                 | 1,958                  | 1,958                  | 1,029          | 1,029                 | 2,071                | 2,071                   | 7,142          | 7,142                   |
| $R^2$                          | 0.38                   | 0.38                   | 0.38                   | 0.39                   | 0.39           | 0.40                  | 0.38                 | 0.38                    | 0.38           | 0.38                    |

Table 6: Impact of Business Expectations on Growth Expectations

*Notes:* Columns 3-10 display estimates for subsamples consisting of firms whose business is independent of the German business cycle, that is firms that report that the German business cycle is unimportant for their business (columns 3-4), firms that make more than 75 % of their sales abroad (columns 5-6), firms with low historic correlation between their business assessment and German GDP growth in the past (columns 7-8), firms in industries uncorrelated with German GDP growth (columns 9-10). \*p<0.1, \*\*p<0.05, \*\*\*p<0.01. Standard errors clustered at county level in parentheses.

that when firms see their business situation to be more favorable they tend to report 0.3 percentage points higher growth expectations. The size of a firm does not turn out to be relevant (column 4).

Columns 3-10 of Table 5 and Table 6 display the results for the second strategy. When we do not differentiate by firm size, the coefficient on business expectations is positive and significant in all subsamples when considering business assessment (Table 5). In the case of business expectations we find significant coefficients only in the low importance and low GDP correlation subsamples. Again, we observe large heterogeneity when we differentiate by firm size. The positive baseline coefficient in each subsample reveals that extrapolation from the business situation is clearly driven by small firms. Except for the foreign sales subsample in Table 6, the baseline coefficients are significant at least at the 5 % level with similar levels in both tables. In the case of business assessment they range from 0.55 for firms in acyclical industries (column 10) to 3.3 for firms with high levels of foreign sales (column 6). For larger firms this effect is either considerably weaker or non existent. Nearly all interaction coefficients have negative signs and yield an overall zero effect for the larger size categories.

|                                 |         |         | All firms |         |            | Acyclic  | al firms          |
|---------------------------------|---------|---------|-----------|---------|------------|----------|-------------------|
|                                 |         |         |           |         | w/o 2020   | (low GDP | corr. $q_{0.1}$ ) |
|                                 | (1)     | (2)     | (3)       | (4)     | (5)        | (6)      | (7)               |
| Price exp.                      | 0.21*** | 0.19*** | 0.22***   | 0.49*** | 0.05       | -0.21    | 0.40              |
|                                 | (0.05)  | (0.05)  | (0.07)    | (0.15)  | (0.062)    | (0.27)   | (0.68)            |
| Price exp. $\times$ Firm size 2 |         |         |           | -0.37** | 0.01       |          | -0.55             |
|                                 |         |         |           | (0.18)  | (0.07)     |          | (0.78)            |
| Price exp. $\times$ Firm size 3 |         |         |           | -0.10   | 0.09       |          | -0.34             |
|                                 |         |         |           | (0.22)  | (0.09)     |          | (0.95)            |
| Price exp. $\times$ Firm size 4 |         |         |           | -0.40   | -0.12      |          | -0.83             |
|                                 |         |         |           | (0.25)  | (0.11)     |          | (1.15)            |
| Price exp. $\times$ Firm size 5 |         |         |           | -0.38   | 0.15       |          | -1.25             |
|                                 |         |         |           | (0.24)  | (0.10)     |          | (0.99)            |
| Wave FE                         | Yes     | Yes     | Yes       | Yes     | Yes        | Yes      | Yes               |
| Region FE                       | No      | Yes     | No        | No      | No         | No       | No                |
| Industry FE                     | No      | Yes     | No        | No      | No         | No       | No                |
| Firm FE                         | No      | No      | Yes       | Yes     | Yes        | Yes      | Yes               |
| Controls                        | No      | Yes     | No        | No      | No         | No       | No                |
| N                               | 33,493  | 28,124  | 33,493    | 33,493  | $23,\!656$ | 2,050    | 2,050             |
| $R^2$                           | 0.30    | 0.34    | 0.38      | 0.38    | 0.09       | 0.37     | 0.37              |

Table 7: Impact of Price Expectations on Growth Expectations

*Notes:* Columns 6-7 display estimates for a subsample of firms with low historic correlation between their business assessment and German GDP growth in the past. \*p<0.1, \*\*p<0.05, \*\*\*p<0.01. Standard errors clustered at county level in parentheses.

Lastly, we consider expectations about individual product prices as a proxy for local signals. For the full sample, we find a positive correlation between those price expectations and firms' growth expectations that is significant at the 1 % level (Table 7). A firm that expects its prices to go up in the next months, on average, reports 0.2 percentage points higher growth expectations (column 3). Regarding firm size heterogeneity, we observe a similar pattern as in previous examples: the effect is mainly driven by small firms. For larger firms the overall effect is not significantly different from zero (except for the case of firm size category 3). Furthermore, we do not find a significant correlation between product price expectations and growth expectations for the pre-COVID-19 sample and the subsample of firms whose business situation was uncorrelated with German GDP growth in the past—even though the signs of coefficients are unchanged in the latter case (column 7).

Overall, the results in this subsection indicate that firms seem to extrapolate systematically from their business situation when forming expectations about aggregate growth. Considering the sample elicited before 2020, the inclusion of business expectations or business assessment leads to an increase in  $R^2$  by 34 % and 29 % compared to a model with only wave-target year fixed effects. Considering both regressors jointly leads to an increases by as much as 48 %. In contrast, product price expectations do not explain much of the variation in growth expectations: including them increases the share of explained variation in growth expectations by only about 5 %.

#### 3.3 Consistency with Theoretical Models of Expectation Formation

We now discuss to what extent the three empirical facts about growth expectations of firms that we have presented so far are consistent with the predictions of the main types of theoretical models with imperfect information structures that are discussed in the macroeconomic literature. We focus on the "island" model by Lucas (1972), the sticky information model with endogenous updating choice as in Reis (2006), the model of Maćkowiak and Wiederholt (2009) with limited attention capacity which builds on Sims (2003), and the model with "sentiment shocks" of Angeletos and La'O (2013). We summarize the consistency between models and empirical facts in Table 8.

The first empirical fact is that expectations are highly dispersed. All four models are able to replicate this fact because different agents—at each point in time—form their expectations about the future aggregate state of the economy based on different information sets.

The second empirical fact is that the dispersion of expectations is larger for small firms than for large firms. The models by Lucas (1972) and Angeletos and La'O (2013) cannot explain such heterogeneity across the firm size distribution because they imply exogenous mechanisms of information provision that are independent of the size of firms. Hence, the nature of information flows is independent of the size of a firm which rules out a systematic relationship between the degree of expectation dispersion and firm size. In contrast, information acquisition is endogenous in the other two models. In both cases, the model structure implies that it is optimal for large firms to acquire information about the state of the aggregate economy more frequently or to a greater extent relative to small firms—leading to less dispersed expectations among large firms. In the model by Reis (2006), firms need to pay a fixed cost to update their information set; the payoff

|                                 | Dispersion   | Size<br>Dependence | Local<br>Information   |
|---------------------------------|--------------|--------------------|------------------------|
| Lucas (1972)                    | $\checkmark$ | _                  | $\checkmark^{\dagger}$ |
| Reis (2006)                     | $\checkmark$ | $\checkmark$       | _                      |
| Maćkowiak and Wiederholt (2009) | $\checkmark$ | $\checkmark$       | √§                     |
| Angeletos and La'O (2013)       | $\checkmark$ | _                  | *                      |

Table 8: Consistency of Theoretical Models and Empirical Results

*Notes:* <sup>†</sup> The model implies additional confusion between nominal and real shocks. <sup>§</sup> Extended version of the model where firms can choose to observe linear-combinations of aggregate and idiosyncratic signals. <sup>\*</sup> Model features noisy idiosyncratic signals about local conditions in other parts of the economy instead of information about a firm's own local environment.

from being able to adjust prices optimally is larger for firms with large sales volumes. Therefore, large firms will update their information sets more frequently. In one of the model setups discussed in Maćkowiak and Wiederholt (2009, Section VII.B.), firms choose how much attention to pay to aggregate conditions, idiosyncratic information, and/or a linear combination of both. Maćkowiak and Wiederholt (2009) show that firms only pay substantial attention to aggregate conditions when the capacity for absorbing information is large. Presumably, this is the case only for large firms. In turn, small firms need to infer expectations about aggregates from the mixed signal that is influenced by idiosyncratic information—and, thus, have more dispersed expectations. In contrast, larger firms also pay attention to aggregate information that is the same for all firms—leading to smaller degree of expectation dispersion.

The third empirical fact is that "local" information influences expectations of firms about aggregate growth. As just described, this is a property of the model by Maćkowiak and Wiederholt (2009) when firms infer future aggregate conditions from a signal that is a linear combination of idiosyncratic and aggregate conditions. In contrast, it is clear that the model in Reis (2006) is not consistent with this fact. When firms update their information set in this model, they always obtain full information about the state of the world; dispersion in this model arises due to outdated information sets rather than any confusion of local and aggregate information. Things are more complicated in case of the two other models. In Lucas (1972), firms indeed have to infer the aggregate state of the economy from local signals. More precisely, however, they observe local prices (a nominal variable) and have to infer aggregate demand (a real variable). So in addition to having to infer something aggregate from local information they face the problem of distinguishing monetary from real shocks. Most of our empirical setups focus on local *real* conditions and, hence, do not fully match this theoretical framework. We provide only tentative evidence in support of Lucas' model in form of the correlations between individual product price expectations and firms' growth expectations documented in the previous subsection. In Angeletos and La'O (2013)—a model that features "islands" as in Lucas (1972), with imperfect information flows between those "islands"—firms receive (noisy) signals about fundamentals and information sets of one trading partner ("one other island"). Hence, their expectations about future aggregates are driven by information about conditions in *other* parts of the economy. This is not in line with our understanding of local information, which implies that firms gain information about *their own* local environment.

In sum, while all four models are consistent with dispersed macroeconomic expectations of firms, only a version of the model framework with limited information processing capacities in Maćkowiak and Wiederholt (2009) can account for all three empirical facts that we describe above. The other theoretical models either fail to generate the negative relationship between expectation dispersion and firm size or do not provide a mechanism by which information about local/idiosyncratic conditions influence macroeconomic expectations. In all models, expectations determine decisions of firms. We analyze empirically whether growth expectations of firms correlate with subsequent business decisions in the next section.

#### 4 Influence on Firm Behavior

Eventually, it is of interest whether heterogeneous expectations lead to differences in firm behavior. The BTS allows us to analyze the correlation between growth expectations and firm responses to questions about employment and investment that are regularly included in the survey.

|                                  |          | Forward       | l looking     |               | Ba            | ckward look   | ing            |
|----------------------------------|----------|---------------|---------------|---------------|---------------|---------------|----------------|
|                                  |          |               |               | w/o $2020$    |               |               | $\rm w/o~2020$ |
|                                  | (1)      | (2)           | (3)           | (4)           | (5)           | (6)           | (7)            |
| Growth exp.                      | 0.000    | 0.000         | -0.002**      | -0.008**      | $0.002^{**}$  | -0.002        | -0.001         |
|                                  | (0.001)  | (0.001)       | (0.001)       | (0.004)       | (0.001)       | (0.001)       | (0.004)        |
| Growth exp. $\times$ Firm size 2 |          |               | 0.001         | $0.022^{***}$ |               | $0.003^{**}$  | 0.008          |
|                                  |          |               | (0.001)       | (0.005)       |               | (0.001)       | (0.005)        |
| Growth exp. $\times$ Firm size 3 |          |               | $0.006^{***}$ | 0.007         |               | $0.007^{***}$ | $0.015^{**}$   |
|                                  |          |               | (0.002)       | (0.007)       |               | (0.002)       | (0.007)        |
| Growth exp. $\times$ Firm size 4 |          |               | $0.006^{***}$ | $0.026^{***}$ |               | $0.005^{**}$  | 0.013          |
|                                  |          |               | (0.002)       | (0.008)       |               | (0.002)       | (0.009)        |
| Growth exp. $\times$ Firm size 5 |          |               | $0.010^{***}$ | $0.042^{***}$ |               | $0.006^{***}$ | $0.034^{***}$  |
|                                  |          |               | (0.002)       | (0.010)       |               | (0.002)       | (0.011)        |
| Bus. exp.                        | 0.199*** | $0.146^{***}$ | $0.147^{***}$ | 0.170***      | 0.049***      | 0.050***      | 0.046***       |
| -                                | (0.004)  | (0.005)       | (0.005)       | (0.006)       | (0.005)       | (0.005)       | (0.007)        |
| Bus. ass.                        | 0.229*** | $0.197^{***}$ | $0.195^{***}$ | $0.142^{***}$ | $0.104^{***}$ | 0.103***      | $0.075^{***}$  |
|                                  | (0.004)  | (0.005)       | (0.005)       | (0.007)       | (0.005)       | (0.005)       | (0.007)        |
| Wave FE                          | Yes      | Yes           | Yes           | Yes           | Yes           | Yes           | Yes            |
| Region FE                        | No       | No            | No            | No            | No            | No            | No             |
| Industry FE                      | No       | No            | No            | No            | No            | No            | No             |
| Firm FE                          | No       | Yes           | Yes           | Yes           | Yes           | Yes           | Yes            |
| Controls                         | No       | No            | No            | No            | No            | No            | No             |
| N                                | 33,222   | 33,222        | 33,222        | $23,\!437$    | 28,320        | 28,320        | 20,266         |
| $R^2$                            | 0.22     | 0.14          | 0.14          | 0.10          | 0.04          | 0.04          | 0.03           |

Table 9: Employment and Growth Expectations

Notes: \*p<0.1, \*\*p<0.05, \*\*\*p<0.01. Standard errors clustered at county level in parentheses.

We regress those measures of investment and employment changes on growth expectations —controlling for business expectations and assessment and the usual fixed effects and control variables (Tables 9 and 10). The first four columns in both tables show the results for the forward looking measures, the final three columns show the results for the backward looking measures.

The estimates of the coefficient corresponding to growth expectations are not significantly different from zero in the case of employment, no matter which fixed effects and controls are included (columns 1 and 2). We do find an effect, however, when we consider firm size heterogeneity in column 3. Interestingly, the coefficients on the interaction terms increase with firm size which implies that large firms base plans for future employment decisions on growth expectations. The same relationship with even larger coefficients holds for the sample elicited before the pandemic. Yet, the coefficients suggest an economically small effect.

The final columns of Table 9 display the results for backward-looking measures of employment. We find a similar pattern as in the forward looking case, both for the full

|                                  |               | Forward    | l looking  |                | Ba          | ckward loo  | king          |
|----------------------------------|---------------|------------|------------|----------------|-------------|-------------|---------------|
|                                  |               |            |            | $\rm w/o~2020$ |             |             | w/o 2020      |
|                                  | (1)           | (2)        | (3)        | (4)            | (5)         | (6)         | (7)           |
| Growth exp.                      | $0.002^{**}$  | 0.001      | 0.001      | 0.000          | 0.002       | 0.001       | -0.001        |
|                                  | (0.001)       | (0.001)    | (0.002)    | (0.007)        | (0.001)     | (0.003)     | (0.007)       |
| Growth exp. $\times$ Firm size 2 |               |            | 0.001      | 0.003          |             | 0.001       | 0.003         |
|                                  |               |            | (0.002)    | (0.009)        |             | (0.003)     | (0.010)       |
| Growth exp. $\times$ Firm size 3 |               |            | -0.002     | $0.033^{***}$  |             | 0.004       | -0.015        |
|                                  |               |            | (0.003)    | (0.011)        |             | (0.004)     | (0.013)       |
| Growth exp. $\times$ Firm size 4 |               |            | 0.003      | $0.026^{*}$    |             | 0.001       | $0.037^{**}$  |
|                                  |               |            | (0.004)    | (0.015)        |             | (0.005)     | (0.015)       |
| Growth exp. $\times$ Firm size 5 |               |            | -0.000     | 0.009          |             | -0.002      | $0.062^{***}$ |
|                                  |               |            | (0.003)    | (0.020)        |             | (0.004)     | (0.022)       |
| Bus. exp.                        | 0.144***      | 0.086***   | 0.086***   | 0.050***       | $0.017^{*}$ | $0.017^{*}$ | -0.016        |
|                                  | (0.007)       | (0.008)    | (0.008)    | (0.011)        | (0.010)     | (0.010)     | (0.012)       |
| Bus. ass.                        | $0.154^{***}$ | 0.100***   | 0.100***   | 0.083***       | -0.020*     | -0.020*     | 0.008         |
|                                  | (0.006)       | (0.009)    | (0.009)    | (0.012)        | (0.011)     | (0.011)     | (0.013)       |
| Wave FE                          | Yes           | Yes        | Yes        | Yes            | Yes         | Yes         | Yes           |
| Region FE                        | No            | No         | No         | No             | No          | No          | No            |
| Industry FE                      | No            | No         | No         | No             | No          | No          | No            |
| Firm FE                          | No            | Yes        | Yes        | Yes            | Yes         | Yes         | Yes           |
| Controls                         | No            | No         | No         | No             | No          | No          | No            |
| N                                | 25,860        | $25,\!860$ | $25,\!860$ | 18,105         | $21,\!182$  | $21,\!182$  | 17,002        |
| $R^2$                            | 0.06          | 0.02       | 0.03       | 0.02           | 0.06        | 0.06        | 0.08          |

Table 10: Investment and Growth Expectations

Notes: \*p<0.1, \*\*p<0.05, \*\*\*p<0.01. Standard errors clustered at county level in parentheses.

sample and the sample elicited before the pandemic. The larger a firm, the higher the degree to which changes in aggregate expectations are able to explain reported subsequent employment adjustments.

The results for the investment decisions are less clear-cut. In the case of the forwardlooking measure, the coefficients have the expected sign but the estimate is significantly different from zero only without firm fixed effects in column 1. Firm size does not seem to play a role in the full sample (column 3). We observe some effects on investment plans for larger firms when excluding the 2020 survey (column 4). Similarly, when using the backward looking measure only estimates based on the sample without the pandemic survey wave suggest that large firms adjust investment in response to prior changes in their aggregate expectations (column 6).

Overall, we find that especially large firms seem to take their growth expectations into account when making managerial decisions (about employment and investment). We interpret this as a sign that firms might be aware of the quality of their expectations. Small firms know that their growth expectations are relatively uninformed. As a result it is rational for them not to base decisions on those expectations. Large firms, in contrast, treat their growth expectations as valuable information, especially before the COVID-19 pandemic.

#### 5 Conclusion

In recent years, macroeconomists have increasingly tried to understand how private households and firms form expectations about macroeconomic variables. Based on a large and fairly representative sample of German firms, this paper shows that local information observed by firms in their industry, their region, or simply within the firm—influences the firms' expectations of aggregate growth.

Overall, our results highlight the importance of idiosyncratic and local information shocks to individual firms for understanding the dynamics of macroeconomic expectations. In particular, they offer an explanation for the observed cross-sectional expectation disagreement across firms. In total, local unemployment, industry growth, business expectations, business assessment, and price expectations (our measures of local information) increase the share of variation in expectations that we can explain by about 45 %. Replacing unemployment and industry growth with county and industry fixed effects even leads to an almost threefold increase in the explained variance share.

Our evidence complements findings in Andrade et al. (2022): our analysis covers not only the manufacturing sector but also the trade and service sectors, it is based on quantitative expectations of aggregate growth, and—due to our knowledge of the location of firms' headquarters—we can link expectations about aggregates to the regional economic situation rather than only to industry conditions. In addition, we provide evidence that the extrapolation from local information to expectation about aggregate growth is more pronounced for small firms than for large firms.

A drawback of our data set is the very small time dimension. Data on growth expectations is only available from four survey waves and for four target years. This precludes any analysis of dynamic effects like the ones shown in Andrade et al. (2022) and confines our study to an analysis of cross-sectional variation in the data. Our results have implications for theoretical modeling of expectation heterogeneity in macroeconomic models. All four models that we consider (Lucas, 1972; Reis, 2006; Maćkowiak and Wiederholt, 2009; Angeletos and La'O, 2013) are consistent with dispersed macroeconomic expectations of firms. But only one version of the model framework with limited information processing capacities in Maćkowiak and Wiederholt (2009) is additionally consistent with dispersion being larger for small firms and an impact of local information shocks on expectations about aggregate variables. The other three models miss to explain at least one of those two additional empirical facts.

Going one step further, we provide evidence that growth expectations are related to firms' decisions about employment and investment, complementing evidence in Coibion et al. (2020). The effect is driven by larger firms and not visible for small firms. This suggest that small firms are aware of the limitations of their macroeconomic expectations and do not base their business decisions on them. Again, this speaks for models of rational expectations with information frictions rather than unconscious expectation biases.

From the perspective of firms, an overreaction to local information could lead to substantial misguided business decisions. We leave it to future research to analyze if firms that extrapolate from local information when forming growth expectations and, at the same time, react to these growth expectations when making employment and investment decisions suffer in terms of business performance in the medium term.

#### References

- Andrade, P., Coibion, O., Gautier, E., Gorodnichenko, Y., 2022. No Firm Is an Island? How Industry Conditions Shape Firms' Expectations. Journal of Monetary Economics 125, 40–56.
- Angeletos, G.-M., La'O, J., 2013. Sentiments. Econometrica 81 (2), 739–779.
- Bachmann, R., Elstner, S., 2015. Firm Optimism and Pessimism. European Economic Review 79, 297–325.
- Bachmann, R., Zorn, P., 2020. What Drives Aggregate Investment? Evidence From German Survey Data. Journal of Economic Dynamics and Control 115, 103873.
- Beaudry, P., Portier, F., 2007. When Can Changes in Expectations Cause Business Cycle Fluctuations in Neo-classical Settings? Journal of Economic Theory 135 (1), 458–477.

- Berger, H., Ehrmann, M., Fratzscher, M., November 2009. Forecasting ECB Monetary Policy: Accuracy Is a Matter of Geography. European Economic Review 53 (8), 1028– 1041.
- Berger, H., Ehrmann, M., Fratzscher, M., 2011. Geography, Skills or Both: What Explains Fed Watchers' Forecast Accuracy of US Monetary Policy? Journal of Macroeconomics 33 (3), 420 – 437.
- Bruine De Bruin, W., Manski, C. F., Topa, G., van der Klaauw, W., 2011. Measuring Consumer Uncertainty about Future Inflation. Journal of Applied Econometrics 26 (3), 454–478.
- Candia, B., Coibion, O., Gorodnichenko, Y., May 2021. The Inflation Expectations of U.S. Firms: Evidence From a New Survey. Working Paper 28836, National Bureau of Economic Research.
- Candia, B., Coibion, O., Gorodnichenko, Y., May 2022. The Macroeconomic Expectations of Firms. Working Paper 30042, National Bureau of Economic Research.
- Coibion, O., Gorodnichenko, Y., 2015. Information Rigidity and the Expectations Formation Process: A Simple Framework and New Facts. American Economic Review 105 (8), 2644–2678.
- Coibion, O., Gorodnichenko, Y., Kumar, S., 2018. How Do Firms Form Their Expectations? New Survey Evidence. American Economic Review 108 (9), 2671–2713.
- Coibion, O., Gorodnichenko, Y., Ropele, T., 2020. Inflation Expectations and Firm Decisions: New Causal Evidence. The Quarterly Journal of Economics 135 (1), 165–219.
- Das, S., Kuhnen, C. M., Nagel, S., 2020. Socioeconomic Status and Macroeconomic Expectations. The Review of Financial Studies 33 (1), 395–432.
- Dovern, J., Fritsche, U., Slacalek, J., 2012. Disagreement among Forecasters in G7 Countries. Review of Economics and Statistics 94 (4), 1081–1096.
- Kuchler, T., Zafar, B., 2019. Personal Experiences and Expectations about Aggregate Outcomes. The Journal of Finance 74 (5), 2491–2542.
- Kumar, S., Afrouzi, H., Coibion, O., Gorodnichenko, Y., 2015. Inflation Targeting Does Not Anchor Inflation Expectations: Evidence from Firms in New Zealand. Brookings Papers on Economic Activity 2015 (2), 151–225.
- Link, S., 2020. The Harmonization of the ifo Business Survey's Micro Data. Journal of Economics and Statistics 240 (4), 543–555.
- Link, S., Peichl, A., Roth, C., Wohlfart, J., 2021. Information Frictions among Firms and Households. CESifo Working Paper 8969.
- Lorenzoni, G., 2009. A Theory of Demand Shocks. American Economic Review 99 (5), 2050–2084.
- Lucas, R. E., 1972. Expectations and the Neutrality of Money. Journal of Economic Theory 4 (2), 103–124.

- Maćkowiak, B., Wiederholt, M., 2009. Optimal Sticky Prices under Rational Inattention. American Economic Review 99 (3), 769–803.
- Malmendier, U., Nagel, S., 2011. Depression Babies: Do Macroeconomic Experiences Affect Risk Taking? The Quarterly Journal of Economics 126 (1), 373–416.
- Malmendier, U., Nagel, S., 2016. Learning from Inflation Experiences. The Quarterly Journal of Economics 131 (1), 53–87.
- Mankiw, N. G., Reis, R., 2002. Sticky Information versus Sticky Prices: A Proposal to Replace the New Keynesian Phillips Curve. The Quarterly Journal of Economics 117, 1295–1328.
- Mankiw, N. G., Reis, R., Wolfers, J., 2003. Disagreement on Inflation Expectations. NBER Macroeconomics Annual 18, 209–248.
- Manski, C. F., 2004. Measuring Expectations. Econometrica 72, 1329–1376.
- Manski, C. F., 2018. Survey Measurement of Probabilistic Macroeconomic Expectations: Progress and Promise. NBER Macroeconomic Annual 2017 32, 411–471.
- Nimark, K. P., 2014. Man-bites-dog Business Cycles. American Economic Review 104 (8), 2320–2367.
- Reis, R., 2006. Inattentive Producers. The Review of Economic Studies 73 (3), 793–821.
- Sauer, S., Wohlrabe, K., 2019. CEO or Intern Who Actually Answers the Questionnaires in the ifo Business Survey? CESifo Forum 20 (2), 29–31.
- Sims, C., 2003. Implications of Rational Inattention. Journal of Monetary Economics 50 (3), 665–690.
- Tanaka, M., Bloom, N., David, J. M., Koga, M., 2020. Firm Performance and Macro Forecast Accuracy. Journal of Monetary Economics 114, 26–41.
- Woodford, M., 2003. Imperfect Common Knowledge and the Effects of Monetary Policy. In: Aghion, P., Frydman, R., Stiglitz, J., Woodford, M. (Eds.), Knowledge, Information and Expectations in Modern Macroeconomics: In Honor of Edmund S. Phelps. Princeton University Press, Princeton, pp. 25–58.

#### Appendix A Figures and Tables

Figure A.1: Dispersion of Growth Expectations for Different Firm Sizes and Levels of Importance of the German Business Cycle









Notes: The graphs show the standard deviation of firms' GDP expectations according to the five firm size categories and their assessment of how important the general economic situation in Germany is for their own business from 1 (important) to 5 (unimportant).

| Survey        | Aug '18   | Mar       | : '19     | Aug       | g '19     | Aug       | g '20     |
|---------------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|
| Expectation   | 2018      | 2019      | 2020      | 2019      | 2020      | 2020      | 2021      |
| Manufacturing |           |           |           |           |           |           |           |
| Ν             | $1,\!541$ | $1,\!664$ | $1,\!649$ | 1,759     | 1,753     | $1,\!806$ | 1,788     |
| Mean          | 1.9       | 1.2       | 1.1       | 0.7       | 0.7       | -6.7      | 1.8       |
| Median        | 1.8       | 1.2       | 1.0       | 0.9       | 1.0       | -8.0      | 3.0       |
| Std. Dev.     | 1.3       | 1.5       | 1.5       | 1.5       | 1.8       | 7.0       | 5.6       |
| Trade         |           |           |           |           |           |           |           |
| Ν             | $1,\!396$ | $1,\!447$ | $1,\!431$ | $1,\!386$ | $1,\!374$ | $1,\!401$ | $1,\!390$ |
| Mean          | 1.7       | 1.2       | 1.0       | 1.0       | 0.8       | -4.5      | 1.5       |
| Median        | 1.7       | 1.2       | 1.0       | 1.0       | 1.0       | -6.0      | 2.0       |
| Std. Dev.     | 1.1       | 1.3       | 1.4       | 1.5       | 1.7       | 8.4       | 6.0       |
| Services      |           |           |           |           |           |           |           |
| Ν             | 1,704     | 1,720     | $1,\!694$ | 1,711     | 1,704     | 1,803     | 1,783     |
| Mean          | 1.9       | 1.3       | 1.3       | 1.1       | 0.9       | -7.0      | 1.4       |
| Median        | 1.8       | 1.3       | 1.1       | 1.0       | 1.0       | -8.0      | 2.0       |
| Std. Dev.     | 1.5       | 1.4       | 1.5       | 1.6       | 1.7       | 8.3       | 6.8       |

Table A.1: Descriptive Statistics for Quantitative Growth Expectations

Notes: All numbers refer to the entire sample after dropping any observations that exceed the median by three standard deviations in each wave. In total, we drop about 1 % of all answers.

|                                 | Ν         | Mean  | Median | SD    | Min   | Max    |
|---------------------------------|-----------|-------|--------|-------|-------|--------|
| GDP importance                  | 5,302     | 2.13  | 2      | 0.90  | 1.00  | 5.00   |
| Foreign sales                   | 4,963     | 15.72 | 3      | 23.78 | 0.00  | 100.00 |
| Unemployment (MA)               | 16,974    | 5.07  | 5      | 2.14  | 1.29  | 13.94  |
| Industry growth                 | 16,917    | 0.03  | 0      | 0.76  | -2.80 | 4.27   |
| Bus. exp.                       | 19,311    | -0.03 | 0      | 0.65  | -1.00 | 1.00   |
| Bus. ass.                       | 19,326    | 0.20  | 0      | 0.71  | -1.00 | 1.00   |
| Investment forward              | 14,966    | 0.05  | 0      | 0.72  | -1.00 | 1.00   |
| Investment backward             | 6,994     | -0.03 | 0      | 0.75  | -1.00 | 1.00   |
| Employment forward              | 19,191    | 0.04  | 0      | 0.52  | -1.00 | 1.00   |
| Employment backward             | 16,382    | 0.02  | 0      | 0.46  | -1.00 | 1.00   |
| Firm size                       | $7,\!401$ | 2.39  | 2      | 1.20  | 1.00  | 5.00   |
| Dummy online                    | 19,407    | 0.79  | 1      | 0.41  | 0.00  | 1.00   |
| Hist. vol. business exp.        | $7,\!390$ | 0.52  | 1      | 0.16  | 0.00  | 1.41   |
| Success predicting own business | $6,\!600$ | 0.27  | 0      | 0.24  | 0.00  | 1.17   |
| GDP - bus. ass. correlation     | $4,\!170$ | 0.04  | 0      | 0.08  | 0.00  | 0.49   |
|                                 |           |       |        |       |       |        |

 Table A.2: Descriptive Statistics for Other Variables

**Notes**: We calculate all descriptive statistics for each firm in each survey wave used in the analysis. Missing values are left out to ensure anonymity of firms.

|  | Firm size 1    | Firm size 2    | Firm size 3    | Firm size 4 |
|--|----------------|----------------|----------------|-------------|
| Firm size 2  | 0.8403***      |                |                |             |
| Firm size 3  | $0.6983^{***}$ | $0.8310^{**}$  |                |             |
| Firm size 4  | $0.5761^{***}$ | $0.6857^{***}$ | $0.8251^{***}$ |             |
| Firm size 5  | $0.4729^{***}$ | $0.5628^{***}$ | $0.6773^{***}$ | 0.8209***   |
| Levene's statistic for the equality of variances between groups: 59.184*** |                |                |                |             |

Table A.3: Differences in Variances - Pooled Sample

*Notes*: This table shows the results of variance ratio tests comparing the variances of growth expectations between two firm size groups and the Levene's test for the equality of variances between all firm size groups. The values in the upper panel of the table refer to the variance ratio. A variance ration smaller than one implies expectations of larger firms are less dispersed.

GDP imp. 1 GDP imp. 2 GDP imp. 3 GDP imp. 4 GDP importance 2  $0.9484^{**}$ GDP importance 3 1.0596\*\*\* 1.0050GDP importance 4 0.95611.0081 0.9513GDP importance 5  $1.3495^{***}$ 1.4229\*\*\* 1.3428\*\*\* 1.4115\*\*\*

Table A.4: Differences in Variances - Pooled Sample

Levene's statistic for the equality of variances between groups: 2.035\*

*Notes*: This table shows the results of variance ratio tests comparing the variances of growth expectations between two groups of firms with different answers to the GDP importance question and Levene's test for the equality of variances between all groups. The values in the upper panel of the table refer to the variance ratio. A variance ration larger than one implies expectations of that firms that assign more importance to the German business cycle are less dispersed.

#### Appendix B Wording of Questions

The wording of the special questions in the BTS were as follows. To ask about the expectations for the annual growth rate of real GDP in 2018 we asked:

Um wie viel Prozent wird sich Ihrer Einschätzung nach das reale Bruttoinlandsprodukt in Deutschland 2018 im Vergleich zum Vorjahr ändern?

\_\_\_\_%

#### English translation (by authors):

According to your assessment, by how much percent will the real gross domestic product in Germany change in the year 2018 relative to the previous year?

\_\_\_\_%

To ask about the expectations for the annual growth rates of real GDP in 2019 and 2020 we asked:

Um wie viel Prozent wird sich Ihrer Einschätzung nach das reale Bruttoinlandsprodukt in Deutschland in den unten genannten Jahren im Vergleich zum jeweiligen Vorjahr ändern? (Prozentangabe mit einer Nachkommastelle möglich.)

2019: \_\_\_\_% 2020: \_\_\_\_%

#### English translation (by authors):

According to your assessment, by how much percent will the real gross domestic product in Germany change in relation to the respective previous year in the following years? (You can state your answer with one decimal.)

2019: \_\_\_\_% 2020: \_\_\_\_%

To elicit how important the German business cycle is for each firm we asked:

Wie wichtig ist die allgemeine Konjunkturentwicklung in Deutschland für die Geschäftslage Ihres Unternehmens?

 $\Box$  sehr wichtig

 $\Box$  wichtig

- $\Box$  bedingt wichtig
- $\Box$  we niger wichtig
- $\Box$  unwichtig

English translation (by authors):

How important is the aggregate business cycle in Germany for the business situation of your firm?

 $\Box$  very important

 $\Box$  important

 $\Box$  somewhat important

 $\Box$  of small importance

 $\Box$  not important

To elicit the share of a firm's turnover made abroad we asked:

Wie viel Prozent Ihres Umsatzes erzielt Ihr Unternehmen / Betrieb im Ausland?

\_\_\_%

English translation (by authors): Which share of your firm's turnover is made abroad (in percent)?

\_\_\_\_%